Author's response to reviews

Title: Protocol for the Cognitive Interventions and Nutritional Supplements (CINS) trial: A randomized controlled multicenter trial of a brief intervention (BI) versus a BI plus cognitive behavioral treatment (CBT) versus nutritional supplements for patients with long-lasting muscle and back pain

Authors:

Silje E Reme (silje.reme@uib.no)
Torill H Tveito (torill.tveito@uni.no)
Trudie Chalder (trudie.chalder@kcl.ac.uk)
Tormod Bjørkkjær (tbi@nifes.no)
Aage Indahl (Aage.Indahl@rehabilitering.net)
Jens I Brox (jens.ivar.brox@rikshospitalet.no)
Egil Fors (egil.fors@stolav.no)
Eli M Hagen (emhagen@online.no)
Hege R Eriksen (hege.eriksen@uni.no)

Version: 3 Date: 7 July 2011

Author's response to reviews:

Dear Editor,

The following revision has been made to the manuscript:
- The list of authors in the manuscript is now written exactly as they are in the submission system
- Unfortunately we were not able to crop Figure 1 as requested. We therefore kindly ask for your assistance with this.

Thank you.